% SIMIODETemplate-TeacherORStudent.tex and simiode.cls for LaTeX2e
% Version 29 July 2015.
%
% This class (simiode.cls) is a modification of that used by
% the journal PRIMUS - Problems, Resources, and Issues in
% Mathematics Undergraduate Studies.
%
% See http://www.tandfonline.com/loi/upri .
%
% The class was developed by  Marc Frantz with suggestions of the
% PRIMUS Editorial Board, especially John Thoo.
%
% This template is to be used with the  class file "simiode.cls".
%
% The latest version of this class and template files
% can be found at
%
%       <http://www.SIMIODE.org> in the Author section.
%
% A "Quick Start Guide" appears below.
%
% This file must be processed with version LaTeX2e or higher.
%
\documentclass{simiode}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% With this version of the tex file and the associated SIMIODE.cls file you can produce BOTH a Student Version and a Teacher Version of your modeling scenario or technique narrative.
%
% Basically the Student Version contains the title STUDENT VERSION along with the NAME OF SCENARIO, with no abstract, no keywords, no tags, no author identifiers, and no COMMENTS section. The Student Version contains only the STATEMENT of the modeling scenario with appropriate figures, data, instructions, and perhaps a bibliography appropriate for student resourcing, but nothing in the bibliography which would "give away" the nature of the modeling scenario, etc.
%
%The Teacher Version contains the material in the Student Version, namely the title NAME OF SCENARIO, now headed with   TEACHER VERSION, but also full author  identification, abstract, key words, tag words, and, following the STATEMENT, a section of COMMENTS which can be discussion of presentation  ideas, "solutions," techniques, etc.
%
% Authors can prepare a full Teacher Version using the notions immediately below  to create a pdf for review (but in review please eliminate all information about the author as we use a double-blind referee system. Such information will be restored when accepted and published in SIMIODE at www.simiode.org.
%
%***********   Toggle the following line to turn the teacher version on (1) or off (0). ***********
%
%
\def\TeacherEdition{0} % Toggle this line's value.
\newcommand{\teacher}[1]{\ifnum\TeacherEdition=1 #1 \fi}
%
%
% The \teacher{ } command wraps any text that will appear only in the teacher version of
% the document. See this file for examples. This has been done for you in the document below.
% Rename the pdf after you compile each final version.
% This option was created  by Eric Sullivan, Carroll College, Helena MT.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{amsmath,amssymb}
\usepackage{color}
\usepackage{graphicx}
\usepackage{hyperref}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding double spacing %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Manuscripts should be submitted doubled spaced.  This is
% accomplished via the \baselinestretch command below.
% However, you may obtain an idea of how your article will
% look in final form, so that you can better plan your
% presentation.  To see your manuscript formatted very
% similarly to the on-line published format of  SIMIODE,
% comment the following line out by inserting a percent sign % (%) at its beginning.

\renewcommand{\baselinestretch}{1.3}

%%%%%%%%%%%%%%%%%%%%%%%%%%%
%% NOTE regarding macros %%
%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Any macros defined for your paper should be contained in
% the top matter. Likewise, any environment definitions such
% as \newtheorem or \newenvironment should also go in the
% top matter.
%
% Do not \input your macros as separate files within your
% final version because more files make it harder for the
% editors and users to keep track of and/or modify and
% append your paper.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Top matter is above and
% the beginning of the document below.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{document}
\begin{thebibliography}{99}
\bibitem{CaryInstitute2011}  Cary Institute. 2011.  Mast-consuming animals.  Cary Institute of Ecosystem Studies. \href{http://www.caryinstitute.org/science-program/research-projects/pulsed-resources -and-consumer-communities-terrestrial-systems/mast}{http://www.caryinstitute.org/science-program/research-projects/pulsed-resources -and-consumer-communities-terrestrial-systems/mast}. Accessed 18 December 2013.

\bibitem{ClotfelterEtAl2007}  Clotfelter, E. D., A. B. Perdersen, J. A. Cranford, N. Ram, E. A. Snajdr, V. Nolan, Jr. E. D. Ketterson. 2007.  Acorn mast drives long-term dynamics of rodent and songbird populations.  {\it Oecoligia\/}.  154: 493-503.  \href{http:/pedersen.bio.ed.ac.uk/page3/assets/clotfelter\_et\_al\_2007.pdf}{http:/pedersen.bio.ed.ac.uk/page3/assets/clotfelter\_et\_al\_2007.pdf}.  Accessed 20 July 2016.

\bibitem{Ostfeld1996} Ostfeld, R. S.. C. G. Jones, and J. O. Wolff.  1996.  Of mice and mast:  Ecological connections in eastern deciduous forests. {\it BioScience\/}. 46(5): 323-330. \href{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_et\_al\_1996\_BioSci\_46\_323-330.pdf}{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_et\_al\_1996\_BioSci\_46\_323-330.pdf}.  Accessed 21 July 2016.

\bibitem{OstfeldEtAl2001} Ostfeld, R. S., E. M. Schauber, C. D. Canham, F. Keesing, C. G. Jones, and J. O. Wolff. 2001. Effect of Acorn production and mouse abundance on abundance and {\it Borrelia burgdorferi\/} infection prevalence of nymphal {\it Ixodes scapularis\/} ticks. {\it Vector Borne and Zoonotic Disease\/}. 1(4): 55-64.  \href{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_et\_al\_2001\_VBZD\_1\_55-63.pdf}{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_et\_al\_2001\_VBZD\_1\_55-63.pdf}. Accessed 20 July 2016.

\bibitem{OstfeldHolt2004} Ostfeld, R. S. and R. D. Holt. 2003.  Are predators good for your health? Evaluating evidence for top-down regulation of zoonotic disease reservoirs. {\it Frontiers in Ecology and the Environment\/}. 2(1): 13-20. \href{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_and\_Holt\_2004\_Front\_Ecol\_Environ\_2\_13-20.pdf}{http://www.caryinstitute.org/sites/default/files/public/reprints/Ostfeld\_and\_Holt\_2004\_Front\_Ecol\_Environ\_2\_13-20.pdf}. Accessed 21 July 2016.
 
\end{thebibliography}


\newpage
% You only need to change the NAME OF SCENARIO in the next line and delete the footnote. See above for instructions on toggling from Teacher Version to Student Version.
\title{\ifnum\TeacherEdition=1 TEACHER VERSION \else STUDENT VERSION \fi\\ACORNS, RODENTS, AND SNAKES}



\teacher{
    \keywords{Acorns, rodents, snakes, trophic levels, masting}




\tagwords{equilibrium, stability, nonlinear, system, differential equation, model, data, forest, predation}
}


\markboth{Acorns, Rodents, and Snakes}{Acorns, Rodents, and Snakes}




\author{Brian Winkel,
SIMIODE\\
Chardon OH 44024 USA\\
BrianWinkel@simiode.org}

\makeStitlePDFLaTex




\begin{abstract}
Students are given the opportunity to build a three trophic level model in which snakes eat rodents, rodents eat acorns, and acorns grow and occasionally mast. While we do not offer any data for validation of student models here, students are challenged to do reality checks and reasonableness tests while modeling adaptability by adding features which reflect sudden increases in acorn numbers, so called masting of oak trees.
\end{abstract}






\section*{SCENARIO DESCRIPTION}

In forests there are many species of plants and animals. As an example, consider  the temperate forests around West Point NY USA and   the nearby Black Rock Forest where there are dens of northern timber rattle snakes. These snakes live by eating a few rodents per season. The rodent prey  are found principally while the snakes are laying in ambush in sunny areas and wandering about their den area. The rodents eat many things, but a principle diet component is acorns from oak trees.  Oak trees produce acorns in some regularity, but occasionally there will be a season in which the oaks mast, i.e. produce an enormous number of acorns. There are also leaner acorn production years. You may have experienced walking in the woods some years in which there is lots of crunching from stepping on a matte of acorns under foot and other seasons when the forest floor is smooth and almost clean of acorns. This is due to masting.  It is believed that masting is a propagation mechanism. For if the oak trees produced the same (predictable) number of acorns each seasons the rodents who consume them would come to a population level at which there were just enough rodents to eat all the acorns. However, when the acorns mast they get way ahead of the rodents and the acorns (which are the seeds of the oak) germinate, take root, and begin the cycle of growth to the canopy that keeps the oak tree population alive in the woods.


We offer you some activities in which you will be guided to build mathematical models of this three trophic level phenomena: acorns, rodents, and snakes. In activities (1) - (8) you need not concern yourself with masting of acorns.

\subsection*{Activity (1)}
Make a sketch and verbally indicate the relationship between acorns, rodents, and snakes. You may introduce other species and interactions, but we suggest you  concentrate on just these three:  acorns, rodents, and snakes.

 \subsection*{Activity (2)}
Make sketches of plots of what you believe   are acorn numbers in a fixed amount of the forest, rodent population numbers in that same region, and snake population numbers in that same region, say over a 10 year period. Describe what is happening in the forest and how your sketched plots reflect this action description.

 \subsection*{Activity (3)}
Verbally describe the rate of change of  acorn, rodent, and snake populations, over time where $a(t)$ represents the number of acorns in a fixed region at time $t$ in years, $r(t)$ represents the number of rodents in the same fixed region at time $t$ in years, and $s(t)$ represents the number of snakes in the same fixed region at time $t$ in years. What contributes to these changes and in what mathematical expression can we realize these contributions?


\subsection*{Activity (4)}

Write down rate equations for $a'(t)$, $r'(t)$, and $s'(t)$. Describe the variables, the parameters, the terms, the relationships, and the reasons for your arithmetic. Discuss with colleagues your models and defend your terms.


\subsection*{Activity (5)}
Now be prudent about the numbers of acorns, rodents, and snakes in the woods, e.g., there probably are lots of acorns, but few snakes with more rodents than snakes, but not as many rodents as acorns. Then make reasonable estimates of your parameters in your rate equations as well as your initial populations of acorns, rodents, and snakes.

\subsection*{Activity (6)}
One way to test the reasonableness of your model is to find equilibrium values. Remember these organisms have lived together in the forest for a long time so there must be some number for each at which they would stay steady, or at least go a bit above and then a bit below or hover around if no catastrophe occurs.  For your model find the equilibria and determine their nature, either stable or unstable.

\subsection*{Activity (7)}
Run your model forward in time, say for 10 years, i.e., solve your system of differential equations with numbers, and plot the numbers for the three organisms over that period of time. Comment on what you model is doing.

\subsection*{Activity (8)}
Improve your model. You decide what criterion you will use to improve your results.

\subsection*{Activity (9)}
Introduce masting as a sudden large increase in acorns and solve your model, again plotting the solutions and commenting on the reasonableness of your results.  You  determine the time and duration of the masting. Adjust values of the level of masting, run your model (i.e. solve your system of differential equations), plot your solutions,   and see how masting changes the model. Explain whether or not this makes sense to you.
  

\subsection*{Activity (10)}

Seek out some data to give realism to your model, e.g., how many rodents per season will a snake eat?  How  big a snake population might there be? How many acorns per acre, hectare, or square meter are there in a normal season?  (See Figure 3 in \cite{OstfeldEtAl2001} which gives a range of between 2.5 and 29 acorns per $m^2$.) In a masting season? (See \cite{CaryInstitute2011} where we learn, `` Heavy acorn production, which generally occurs once every 2-5 years in deciduous forests, is an example of masting.'') How many acorns does a rodent eat per unit time (or take out of circulation, for you have seen squirrels burying acorns and do you think the squirrel finds every acorn buried in preparation for winter eating?) How many rodents might an acre or hectare  support?  Some of this you can estimate from your own experiences, e.g., you see chipmunks around your campsite in a state park and does that give you some idea of their density in the woods?  Are rodents ``breeders'' and how often? You might find data on northern timber rattlesnakes, their den size, their rodent consumption, their roaming range, their birth rates, etc. See the references below. For  data on acorn masting and rodent abundance over a 20 year period see Figure~\ref{fig:6-21-MastMouseData}. Also we note the following

\begin{quote}
``Acorns are a highly nutritious food source for many species of mammals and birds, including white-footed mice. Mice store acorns for winter consumption, when food sources are generally scarce.

``We have used both long-term monitoring and experimental enhancement of acorn abundance to determine the effects of acorns on mouse populations. Long-term monitoring demonstrates that summertime abundance of mice is strongly correlated with acorn production during the prior fall. Experimental acorn additions indicate more mice survive the winter when acorns are abundant and winter reproduction only occurs when acorns are present. Following acorn mast years, as a result of higher over-winter survival and winter breeding, population growth starts earlier in the season, leading to higher peak densities of mice in summer.''\cite{CaryInstitute2011}

\end{quote}

A nice ''conceptual model of the relationships among key functional units in the oak forests of eastern North America'' is offered in Figure 1 of \cite[p. 324]{Ostfeld1996} 






As more evidence of the strong relationship between acorn population and rodent population we  read in \cite[p. 17]{OstfeldHolt2004} that

\begin{quote}
``\ldots the size of the autumn crop of acorns or
other tree seeds is a strong determinant of rodent density
the following year, often explaining over 80\% of interannual
variation. 
\end{quote} 


Finally, we offer a convincing plot in Figure~\ref{fig:6-21-MastMouseData} depicting the relationship between acorn masting and mouse abundance. Notice that the mouse abundance peak is always one year or season after the acorn mast peak.  And it would be reasonable to notice a comparable period of time between the mouse abundance peak and the snake population peak as more mice means more food and a more supportive environment for snake population growth.

\begin{figure}[!h]
\begin{center}
\includegraphics[width=9.5cm]{6-21-MastMouseData.eps}
\vspace*{-7.0 true cm}

\parbox{9 true cm}{\caption{\label{fig:6-21-MastMouseData} From \cite[p.~497]{ClotfelterEtAl2007} we show a plot of the relationship between acorn masting and mouse populations in a forest.}
}
\end{center}
\end{figure}

\teacher{
\section*{NOTES FOR TEACHERS}

This is a rather open ended set of activities to give students a sense of the broad applications of nonlinear system of differential equation models. We do not offer a ``solution" or even solution strategies, but rather encourage students to collect sources and data beyond what we provide. We found the literature in support of the biological aspects of this scenario to be fascinating and so did our students. At the time we visited this modeling opportunity we could not locate suitable data for verification of any model so we lived in the world of toy models, albeit three nonlinear differential equations representing the change in acorn abundance, rodent population, and snake population. Such representations permitted some ``what if'' analyses, but in the absence of data there could not be  a confident confirmation of the models.

}
  










\end{document}
 
